Author’s response to reviews

Title: A fatal case of recurrent amiodarone-induced thyrotoxicosis after percutaneous tracheostomy

Authors:

Vasilios V Papaioannou (papabil69@vodaphone.net.gr)
Erini E Terzi (terziv@civil.auth.gr)
Christos C Dragoumanis (christosdragoumanis@gmail.com)
Dimitrios D Konstandonis (drkonst@in.gr)
Vasiliki V Theodorou (vassiliki.theodoru@gmail.com)
Ioannis I Pneumatikos (ipnevmat@med.duth.gr)

Version: 2 Date: 3 August 2007

Author’s response to reviews: see over
Dear Mrs Tayabali

According your suggestions we make the following changes

1. Title page was reduced to one page
2. Word Counts were removed from the tile page
3. key words from the abstract were removed
4. An Acknowledgements section was included.
5. A Consent section was included.